Author's response to reviews

Title: Chronic Presentation of Boerhaave's Syndrome

Authors:

Umer F Malik (doctorumer82@yahoo.com)
Ryan Young (rynyng@gmail.com)
Heing D Pham (hdpham5@yahoo.com)
Alisha McCon (mcconali@hotmail.com)
Brenda Shen (shen415@gmail.com)
Richard Landres (rlandres@sjgh.org)
Ahmed Mahmoud (amahmoud@sjgh.org)

Version: 3 Date: 6 January 2010

Author's response to reviews: see over
Cover Letter

Title page for your manuscript that includes contact details and affiliations for all authors—**Attached!!**

The manuscript should also include a statement to this effect in the 'Acknowledgements' section, in follows: "Written consent was obtained from the patient or their relative for publication of study".—**Added**

Please include a 'Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write 'The authors declare that they have no competing interests'.—**Added**

Authors' contributions - Please include an Authors' contributions section before the Acknowledgements and Reference list.—**Included**

Amir AI et al. Scand J Gastroenterol. 2004 May;39(5):418-22. As this report was one of the first recommended a conservative approach with minimal morbidity and mortality it is worthy to be mentioned—**Added**

include A commentary and explanation concerning the evolution of this particular case—**Added**

To review the literature looking for similar cases, and trying to establish some common features in these chronic patients with a Boerhaave Syndrome—**Chronic boerhaave is not known in literature, but what ever I found I added in the literature.**